Author's response to reviews

Title: Reliability and validity of Arabic adaptation of short form of Rapid Estimate of Adult Literacy in Dentistry (AREALD-30) in Saudi Arabia

Authors:

Santosh K Tadakamadla (santosh_dentist@yahoo.com)
Amir H Pakpour (dr.faeq.quadri@gmail.com)
Mir Faeq A Quadri (faeq.ali@yahoo.com)
Aadil Inamdar (aadilinamdar@gmail.com)
Siddig I Abdelwahab (siddigroa@yahoo.com)
Jyothi Tadakamadla (docjyotia@yahoo.co.in)

Version: 2
Date: 21 February 2014

Author's response to reviews: see over
To,

Editorial Correspondence
BMC - Oral Health

Dear Sir/Madam,

I am pleased to submit an original research article titled “Reliability and validity of Arabic adaptation of short form of Rapid Estimate of Adult Literacy in Dentistry (AREALD-30) in Saudi Arabia”

We believe that this manuscript is appropriate for publication as it lies within the scope of the journal. Our manuscript creates a paradigm for future studies on the topic and also be an initiative for improving dental awareness and literacy in the region.

This manuscript has not been published and is not under consideration for publication elsewhere.

We have no conflicts of interest to disclose.
Please let me know of your decision at your earliest convenience.

Thank you for your consideration!

Sincerely yours,

Dr. Quadri

Email:- faeq.ali@yahoo.com